2006-03-31 Shutdown Meeting Friday, March 391-65%/?
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April 2006 updated 3724706 (Draft)
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u]
= Requests:
u]
Linac
Finished Work This Week
v" Cleaned L.E. LINAC Modulators & * Continue Preventative Maintenance on
Rectifiers (LFR1-LRF5) H.E. LINAC Klystron Systems
v" Replace water fittings in Klystron * Debugging LRF7 oscillations
Modulator De-Q'ing circuit *  Modulator Characterization on LRF5
v" Replaced Collimator between L.E. |- Anode Bulk Supply Upgrade
LINAC Tanks 1 & 2 *  NTF Patient Treatment (M,W,&Th)
v" L.E. LINAC Water Maintenance
v" Modified Chopper & Moved Quad
, for future MuCool operations Next Week
Completed McCool Alignment * Continue Preventative maintenance on
v Annual Pre-Acc Safety Trip Test H.E. LINAC Klystron Systems
v' Cleaned Filters on 8 lon Pump PS's ¢ LLRF 9" Line Calibration on LRF4
v’ 45°/90° Magnet PS Maintenance *  Driver Characterization on LRF4
v" LLRF 9" Line Calibration on LRF5 ¢ |- Anode Bulk Supply Upgrade
v Driver Characterization on LRF5 * NTF Patient Treatment (Tu,Th,&F)
O
O
Booster
= Summary:
u]
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Booster Shutdown Work - Week 5

+ Q805/Q806 and Septum Alignment continues
* MI-8 beam stop aligned

* Downstream MI8 dump line assembly

* Kicker Beam Tube installed

* LCW pipe for corrector upgrade nearly done
+ ORBMP magnet/Girder work continues

* ORBMP Power Supply Conditioning

* Tunnel Sump Line installation in progress

Booster Shutdown Work - Week 5

* Booster East 13.8KV switch gear installed
* GMPS #3 Transformer replaced

* Candy Cane PEEK and Tubing upgrade on
gradient magnets

* RF maintenance continues

* 400 MeV Gallery work continues
» Booster West MI8 Tunnel coring completed
* MI-8 conduit arrived and is being installed!
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PEEK Tubing
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Conng at 0805/806
é For Septum Cables (condmts)

13.8 kV work/GMPS #3 Transformer
Replaced
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Pbar

Quad BPM
90% done

4-8GHz Core
10% done

2Uop %01
adAjojoid IIEPIoElS

AP30 Tank Leak
100%done

4 Acc BPMs
AP30 0% done
Deb Quad Shunt:

100% done

Upgrade Deb.
90% done

LMC
50% done

BAR Shutdown Status Mar 31, 2006

&l {AP2 BPM pre-amps | =7

7 Survey Level Run
85% done

Insta op g

40% done

AP1PS
Rotate AP2 BPM 50% done
0%done |4 000000

Install New Target Install Hall Probes ;
APO 0% done ' 5% d :
Install OTR .
0% done
[m]

Main Injector

MI Shutdown report

D. Capista 3/31/06

- Continued installation of the new MI8 line multiwires .
« Continued installation of MI8 collimator.

« Continued installation of a large aperture quad at 222.
« Continued installation of large aperture quad at 608.

« Continued installation of large aperture quad at 620.

« Began installation of large aperture quad at 101

o

o Both quad locations - have had problems with o-ring vacuum seals.
o

o Quad 620
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Work in the Tevatron i

H TEL 2 installation AO Separators A49 H

A0
TEL 1 problems \ / 1.5 m collimator
A48

FO

HLS installation
F,A sectors

BO

Level Run
Done

B-4 Vacuum
work

E-2 house ~
Vacuum work

Valves at E17

/ done

Shimming
Xtal Collimator 6 magnets/day «~— Separator B48
IPM work E0 ~~ At F32-4

Co

EO 133/244 done

HLS D sector

o DO
ul
Vacuum work
« Complete F-4, A-3, E-2
« Underway at B-4 change stands B43-1 B44-1
Should be certified early next week
< Start on D-1 next week un roll D16-1 by 12 mr
« IPM stand to shop for re work
« Bake of Xtal Col done, E-17 Schottkys ended
cooling down
« TEL 2 installation on going
« TEL 1 at IB-2 bad bellows have been cut out and
O replacement underway
ul

B43 B44 and | Ring

B43B44

2600 5000
- 4500
2550

o 2 \ 4000

2500 - “ 350

2450 0w
2m [+ Ba3
2400 R " B4s
2000 |—u—|Ring

—

y
o - i

bt o
500
250
e 0
200 S0
1726006 4:48 12606936 172606 1424 1EB0E19:12 172706 000 1706448 12635

Date:

Plot show as we ramp the Tevatron, we can see the quads move. We can try
to chagne the stands to see if it helps.
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to chagne the stands to see if it helps.

Stand Change at B43-1

it

Vi Tl
o
Al

Single Phase He
Feed lines bellows

Jobs not yet started

» Deep rod monuments
* Un Roll D16-1
* Re align low beta quads at BO

» As found A and E sectors with laser
tracker

* A 17 separator
* TEL 1 installation
* |[PM installation

SY120
= Summary:
o Continue to work on low momentum config for MTest
NuMI
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NuMI - Shutdown — Week 5

Construction Shaft Drip Pan Removed

Drain And Calcite Cleaning Completed at Absorber & Muon Alcoves
Starting in Decay Pipe Passageway

Fall Protection in Pre-Target Near Complete

Aluminum Angle Installed at Edge of Emergency Passageway Wall

Job Planning in Process for Target Hall Desiccant Unit Installation
Initiating Other Actions to Reduce Humidity in Target Hall

Level Runs Starting Next Week
Target Inspection Next Week

Spring Shutdown 2006
MiniBooNE

* Week of April 17:
— Test Horn3.

— LCW will be available
sometime during week of
April 10, bet better to fix
adate than scamble.

— Need enclosure
interlocked.

* Week of May 1:
— Begin alignment work.
— Begin RAW system work.

a
o CDF
= Summary:
o Finsihed cabling hardware event filter.
o Some racks have power problems.
o Cleaning alochol bublers for gas systems
o 4 of 12 HVPS. 2 days behind schedule.
= Requests:
O
° DO

Summary of Fifth Week Activities

Layer O
- Transport North Inner H disk to Si Det for mount repair
- Begin testing HV distribution for Layer O channels

- Mount tables, rails, and optical equipment in gaps in preparation for
installing Layer O mounts on SMT support structure

- Begin installation of Layer O mounts on SMT support structure
Level 1 Calorimeter Trigger

- Continue connecting BLS cables to patch panels and testing
- Continue verifying L1Cal Trigger installation
Level 1 Central Track Trigger

- Continue cabling DFEA2s and begin verification of cabling
Detector Maintenance

- Begin reassembling individual luminosity monitor counters
- Continue calorimeter individual channel recovery

- Upgrades of L1 Muon power supplies in progress

- Continue muon PDT mods to accommodate latency shift

Power outage at DZero on Saturday April 1 from 0700 to 1600
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Layer O Installation Milestones

Current e Milestone
ID TASK NAME Actual S Shutdown Dates
Forecast

2 Beginning of Runllb Tevatron Shutdown 2/23/06 2/27/06 2/27/06
28 | Detector Open, Ready for Access 3/1/06 3/2/06 3/07/06
34 | Runlla Be Beam Pipe Disconnected 3/3/06 3/7/06 3/9/06
47 | H Disks Removed 3/15/06 3/17/06 3/23/06
56 | Runlla Be Beam Pipe Removed 3/21/06 3/27/06 3/31/06
75 | Layer 0 Tooling and Mounts Ready 4/7/06 4/10/06 4/17/06
79 | Layer 0 Installed 4/11/06 4/12/06 4/19/06
91 | Runllb Be Pipe Connected, Layer 0 Cabled 4/25/06 4/26/06 5/03/06
94 | Inner H Disks Re-Installed 5/1/06 5/2/06 5/9/06
99 | Silicon Cold and Ready for Technical Commissioning 5/4/06 5/5/06 5/12/06
103 | Complete Technical Commissioning of Silicon 5/10/06 5/11/06 5/18/06

o 118 | Detector Closed for Tevatron Resumption 6/01/06 6/02/06 6/4/06

o

Alignment

Alignment Schedule for March 31 2006

= Tevatron
» Completed 96% Unrolls

» Completed Elevation run 92% (Drops still needed)

» Completed B48 Separators
»  Scheduled Unrolls (Should complete today)
» Scheduled Deep Rod installation.

= Main Injector

» Completed Alignment MI 608
» Completed Alignment MI 620
» Completed Alignment MI 101
» Schedule Texas Multi-Wires

» Schedule MI 402

Alignment Schedule for March 31 2006

= 8Gev

» Scheduled Continue " The installation of the MI-8 beam dump will require
alignment. Several kickers will be installed, one septa magnet, 1 BPM, 2 Toroids,
beam pipe, 1 quad will be moved and two multiwires."

a P-Bar

» Completed 95% Elevation runs (One Drop needed)

= 400Mev

» Completed Additional OrBump Magnet Alignment on Girder IB2

= Recycler

» Completed 180 Deg. Magnet alignment and beam pipe.

Alignment Schedule for March 31 2006

= MiniBooNe

> Not Scheduled

= Numi

» Schedule Level run between MI-60 to NuMI SR3 (Tgt Hall) aprox 1500 ft.

= Linac
» Completed

=MIPP

> Completed
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Alignment Schedule for March 31 2006
= Add On’s

= Tevatraon
> Completed as found Separators B49
» Scheduled Tevatron Unrolls (Today)

= 400Mev
> Additional OrBump Magnet Alignment on Girder 1B2

[¢]
Progress as of March 31 2006
ID [Task Name March [April [May [June
213212012027 | 306 [ 313 [3/20 [3/27 | 4/3 [ 4104117 [ 4124 511 [ 508 [5/15 [6/22] 5129] 605
1 |ShutDown 2006 E ‘ ‘ 1 53%
2 | DO v » 14%
7| LINAC
9 | MESON TEST = 1)
| sY120 (= L
5| TEVATRON e ‘ W 57%
% | 8GEV | — = L
%8 | Training for incoming techs =P 100%
6| MAININJECTOR € ® 8%
8 | NUMI —= ‘ 9 29%
& | RECYCLER v v 0%
% | MINIBOONE L === K
97 | CDF w 100%
07 | BOOSTER [ 1)
103 | PBAR " 5%
10 | MIPP/ESOT v %
o [ TZ7] 400 MEV = 9 21% |

o Mechanical - Booster 95 LCW Presentation
Booster 95°F LCW System
Safeguards/Improvements
A Progress Report

« Current Status

* Review

« Valve Schematic
« Action Plan

Current Status

« Valves required for flow have been verified open
* Atemporary stand down on all main header
valves is in place.
* LCW System Operational, half capacity
— Booster Flow, Enclosure, Gallery, restricted
— Switchyard and LINAC normal configuration

— Temporary Configuration Control procedure in place,
Lock box

— 1st Draft Call and Approval List
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DRAFT

Y DRAFT

\V] > Booster 95 LCW Flow and
_ Valve Configuration
[ gl | \

Booster Enclosure

East Booster Gallel

_Y.
h
- )
- b
over Transe] Q . MXB—(‘ Touiac
. N [5)

- Lower X-Gallery TRCR
To Switchyard Towor lovel)

West Booster
Gallery

31" oypass

Ball
Revision Date: 3/22/06

Short Term

« Configuration Control/LOTO for normal
shutdowns

« Identify and operation control all main
header valves.

« Procedure for emergency situations

Mid Term

Water relief valves upstairs & downstairs
Surge tanks

Review & Update console graphic page
ACNET Console, Improved
instrumentation pressure/flow

LCW System Configuration training
(operators & water techs)

Rookie books

Tunnel

Sup B 1 y

Magrnet
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Long Term

Piping System Reconfiguration

Add bypass in CUB (FESS) or downstairs
or gallery (AD)

Valve Position Monitors/Indicator

Actuate critical valves for control

Very expensive. Cost versus value, time

labor
]
o ES&H
ES&H

m Safety System Upgrades
0 Data Acquisition System 95% Complete
= 16 of 17 Linux Ethernet Port Replicator (LEPR) chassis installed

0 Booster 50% Complete
= Remove RDF Area Interlocks - Complete
= Booster Dump Move Configuration Changes
m  Completing Documentation to submit for Approval to Operate

0O Main Injector 15% Complete
m  Lower Level Controlled Access Displays Replaced
= Main Injector Coasting Beam System changes
= Lam10 Critical Device Interface Installation

ES&H

m Safety System Upgrades
O P-Bar Source 80% Complete
m Radiation Monitor Card Replacement - Complete
m  Safety System Power Supply Replacement
= Interlock New BSC700 Beam Stop

a Tevatron 50% Complete
®  Moving location of both Tevatron Coasting Beam Gate Valves

m  F-Sector Radiation Monitor Card Replacement

O MT6 Section 1 & 2 50% Complete
®  Enclosure Interlock System Upgrade

0 Safety System Testing 0% Complete
m  All areas need to be re-tested before startup.

ES&H

= Radiation Exposure
0 ~4.4 R total dose for the first 4 weeks of the shutdown.
0O Reminder to get weekly dose cards or electronic dose recording
turned in on Monday'’s.

= Radon Detectors

= No Injuries or Incidences report.

o Roger reminds us to keep Safety in mind.

o Cryo
AD/Cryogenics Dept
Shutdown 2006 Status Report
3/31/2006 (Makara/Olesen)

Cryo Operations:
o Tevatron at LN2 temperature and LN2 usage is stable with vendor deliveries.
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[e]

Helium losses are slightly up at 15 kscf/day (1/2 of normal during Tev
operations).
o« Al, A3, B4, and E2 are at room temperature and isolated with LOTO.
F4 is also at room temperature, has been reconnected by Cryo, and is currently under
normal helium flow with anticipation of Division sign-off and cooldown to LN2
temperature soon; this house can be cooled down to LN2 temperature without the
TEL-1 installed.
E2 was finished by MSD and Cryo has started reconnection procedure.
B4 is nearly completed by MSD and Cryo will start reconnection procedure after their
approval.
A3 is also almost completed by MSD; Cryo needs to epoxy past noted helium leaks on
power leads and spool piece conning tower leaks (5 out of 8) still need addressing.
D1 warmup to room temperature will start today and should be ready for MSD by
Monday night.
o Pbar D60 Debuncher vacuum work has been completed by Pbar Dept, thus
Cryo personnel will restore helium/nitrogen connections and cool down tanks to
LN2 temperature after Division sign-off.
o ACNET application programs relating to FRIG operations are being tested on
Linux consoles.
« MCR monitoring cryo operations.

CHL Maintenance:

o General facility maintenance is in progress.

« CHL Helium Compr “A” cooling tower installation has been completed with
painting in progress. Some minor controls/electrical/instrumentation work is left.

« NRL maintenance has been completed. The IR Booster Compressor was
overhauled and the TSI Compressor Niagara Intercooler heat exchanger
bundles were reinstalled. The IR Booster Compressor is being started today for
plant purification. Estimate return of plant back to service is after the April 3
MSS power outage.

o Helium Coldbox-2 controls upgrade is in progress.

FRIG Maintenance:

« Ring-wide overhauls of wet expanders in progress with 17/24 completed.

« Five IHI cold compressors that were removed for inspection have been
overhauled.

« Various FRIG bldg vacuum maintenance and instrumentation repairs (e.g. lost
charge on VPTs) are in progress, as well as support work on houses at room
temperature and other equipment as things fail even during the shutdown period.

« Various Mycom compressor maintenance: leak hunting and general repairs, FO#
1 compressor replacement was completed, FO#3 motor starter replacement in
progress.

« All 4 Kautzky test stands are complete 4 mini kautzky change manifolds
completed. There are presently 250 poppets in inspection. Kautzky change outs
will begin next week.

o TEL-2 LN2 and LHe jumper U-tubes has been installed in the bypass. Four
flex U-tubes for TEL-2 installation are complete, to be used for TEL-2 final
connection.

Other - Linux Migration
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Other - Linux Migration

to LINUX Migration

ificati:

-

March 3

He Edt Vew Go Sockmaks ook Window Hep

¢e.2 3 0
Back  Forward | Reload
rome | Wsockmarks £Googls Lmcailacrg Lmozlszine L czdev.ors

s
=

Accelerator Division - Accelerator Controls

AD Home | Database | Services | Program Libraries - VMS | Notify Display | Tele

Accelerator Controls Groups Java Controls

» Cenral Services Group * Java Applications

+ FIRUS Hardware  ACNET Console

» Microprocessor Group + DAE Machines

+ Console Software Group. « SDA Homepage

» Networking Group

. and Linkes * Tmportant information about

+ Systems Hardware Java Control System upgrade.
Integrated Engineering .

* latemsted Fnnes Accelerator Controls Information

Hardware Documentation

* CVS
+ CAMAC Modules * The Issues — Accelerator Controls' Wiki
© CAMAC Links + Controls Aschive Information
+ Ervors (internal access only)
* VMS Systems
Linux Port (LCPP) Work * E-Log of work (New)

+ Migration Tutorial (user focus)

‘+ Migration Info (developerfkeeper focused)
* Linuz CVS Respository

+ Mightly Buid Report

‘+ Linug Java Console

+ Schematics

Diagnostics and call-in instructions for controls problems.

Local aceess only

The Picture Board is available.

52 B @

 Controls Wigration - Worilla
Took Window Help

(8 ittt o) 3 -

Ahvome Wbmockmaris £Googe Lmozila.org fncallazne. £ ncadev.crg

Accelerator Controls Applications Migration

‘User Information Link: Eroject Managemen: Link: Archive Links

The Accelerator Di p Control's VMSHVAX to Linux PC's. Broadly speaking, the migration will proceed inthree steps
Phase T The console infrastructure migration carried out by the Control's Department

Phase IT: The control's applications migration. This wil be a division wide activity managed out of the Accelerator Integration Department

Phase I VMS services

The mermbers of the core migration tea are: Peter Kasper (Phase IT Leader), Wally Kissel, and Carl Schumann (Phase I Leader). Department liaisons have also been assigned to
work with the core team.

ional Links

P Tiorial
» Novice Keeper Info
» Help on Linux Clb and User Libraries
© Report a Problem or request help including the Unix consoles or operating systemns

 Linux Console Demo
© The PowerPoint sldes for the demo
© Cookbook procedure for the derro
» Uniz for VMS Users: A briefintroduction to Unix
» Mecca under Linus How to use Mecca on L
Linu: How to starta and Lin terminal
uples: Console Windows: A description of how to use Duplex Consoles
» Fising Linu Java Console: A procedure to fifrestart the Java console served from clid9
 File Shere Files How to use the debugaing file sharing area
© Other VMS Files: Copying non-FSHARE VMS files needed on limix.
» Linuss machines status
 Porting Guidelines
» Conan the Librarian(oldelib help)
» Nightly Buid Results
» Wightly Porter Stas Report
© CVS Repository
» Control System Users List Aschive
‘o Cortrol Suster K eeners List Archs

. SO e@a T ) —o-=ia
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List | applications certified between [Mar 242006 00:00

or hst Pending Certification Requests

and [Mar 30 2006 2400

Application  Deseription Certified on  Linux Usage
113 pa1175-D87 check power supplics status Mar 30 2006 1620 Ohrs
112 sa11s8 Measures Booster Bunch Length Mar 30 2006 16:19 0hrs
111. paz0o0 - C45 Read a single bpm from profile frames.  Mar 30 2006 16:19 Ohrs
110. paosss - D90 Tevatron Quench Log Display Program  Mar 30 2006 1619 Ohes
109 sai022 Electron Cooling losses plot Mar 30 2006 1418 Ohrs
108 pa2023 - B33 ReCyeler Schottiey Monitor Mar 30 2006 16:18 Ohrs
107 pa1610-F113 GPS Status Idar 29 2006 10:47 Ohrs
106. pao?ss - H48 gpm status test (for B0 and DO new  Mar 29 2006 1046 Ohrs
(920
105 gallsz CDF downtime logger plot Mar 29 2006 1043 Ohrs
104 sa1083 ODDMOCD Plot. Tuar 29 2006 10:42 Ohrs
103 pa2103 - B122 Calcs Booster Bucket area. Idar 29 2006 10:42 Ohrs
102 pa1s21-D54 Setting Server Browser/Peeker ar 29 2006 10:42 Ohrs
101. pa1008 - D80+ Dump device database information Mar 29 20060913 1356 5 hrs
program
100, pati7s Clock logger (runs onenslinslot7)  Mar 28 2006 1030 Othrs
99 pa1563 - F25 Refiig Temperature and PSIA Plot Mar 28 2006 1023 Othrs
Program
98 pa1507 - F75  Refrigerator mass flow calculations Mar 28 2006 1028 Ohrs
97. pa1ass - F16 This program starts up the Tevatron cryo Mar 28 2006 10:33 0frs
temperature display.
96 pai1z9s - F24  Cryogenic finite state machine viewer Mar 28 2006 1028 70 14hrs
95 paigi1 -F26 TRANSFORM CONSTANT Mar 28 2006 1027 Ohrs
94 pa15e0 - F27 KEautzky leak bit decode and display Idar 28 2006 1027 Ohrs
93 pa1659 - F74 CHL Flow Calculations dar 27 2006 1035 42.2hrs
92 pa00ss - D37 Curve Fit I program which fits datato  Mar 27 2006 1032 0.2hrs
exps or polynomials
91 pan33s - D64 TEST PROGRAM Mar 27 2006 1031 Ohrs
90 pa194s - 136  Main Injector ADB116 Mhux Switch dar 27 2006 1030 28hrs
8% pa1881 - 563 This application launches ACT scripts.  Mar 27 2006 1028 2hrs
Pending Certification Reque.

Application  Description Requested on  Sensitivity
16. pa075e - D43 Lumberjack Logger List Control Mar 21 2006 1224 Very Sensitve
15 pa151s - D44 Lumberjack plotter Mar 21 2006 09.36 Moderate
14. pai94s - P189 PBAR ADB116 MUX SWITCHES  Mar 15 2006 1406 Moderate

CONTROL
13 pa050s - T14  Save and display temperature and Mar 28 2006 1220 Tnsensitive
humidity from the Gannon detector
12 pa1040 - D133 This i a program to test Mar 24 2006 1443 Insensitive
window_manager and related routines
11. pa1617 - D60 Display Obiect Locks Mar 24 2006 1038 Tnsensitive
10. pa1sas - D57 GAS module alarm dowaload fle DMar 24 2006 1057 Insensitive
displayfedtc
9 pa1710 - D36 This program allows the user to create  Mar 24 2006 1031 Tnsensitive
and edit generic structur
8 pa1002 - D110 test walib reutines Mar 24 2006 0921 Insensitive
7. pa1971 - D134 to display and edi device access list ~ Mar 15 2006 1409 Tnsensitive
6 sal1s1 Spectrum Analyzer Emulation Mar 15 2006 1408 Insensitive
5. 380020 Data Pool Manager Peck SA. Mar 15 2006 1408 Insensitive
4. =20017 SAVERESTORE PROGRESS Mar 15 2006 1408 Tnsensitive
DLAGNOSTICS DISPLAY
3. pa144s - D131 Diagnostic in support of tep/ip data Mar 15 2006 1408 Insensitive
acquisition
2 pa0704 - D35 PLD's cache table peck Mar 15 2006 1407 Tnsensitive
1. pa0oos - D3 Data Pool Manager Usage Pecker DMar 15 2006 1407 Insensitive
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